

Author's response to reviews

Title: The Impact of Bone Morphology on the Outcome of the Pivot Shift Test: A Cohort Study

Authors:

Thomas Branch (doctorbranch@yahoo.com)

Shaun Stinton (stinton@gmail.com)

Adrija Sharma (adrija.sharma@gmail.com)

Frederic Lavoie (fredericlavoie.md@gmail.com)

Christian Guier (guier@jhmail.com)

Philippe Neyret (philippe.neyret@chu-lyon.fr)

Version: 5 Date: 06 Oct 2017

Author's response to reviews:

I have uploaded a letter from Prof. Neyret that further describes the IRB situation related to the study.